Author’s response to reviews

Title: Feasibility of referral to a therapist for assessment of psychiatric problems in primary care – an interview study

Authors:

Agneta Pettersson (agneta.pettersson@gmail.com; agneta.pettersson@sbu.se)
Sonja Modin (sonja.modin@gmail.com)
Henna Hasson (henna.hasson@ki.se)
Ingvar Krakau (ingvar.krakau@ki.se)

Version: 2 Date: 07 Aug 2019

Author’s response to reviews:

Dear Editor,

Thank you for considering our manuscript for publication in BMC Family Practice. The manuscript now has been revised as follows:

Question 1:

In your Ethics approval section of the Declarations, please clarify whether consent obtained from participants was written or verbal. If the consent was verbal please include reasoning for this and state if an ethics committee approved this method of obtaining consent.

Response:

We have added that the participants signed the consent form.

Question 2:

Please also indicate the form of consent in the consent for publication section.

Response:
We have added that the consent for publication was written.

Question 3:
In accordance with our submission guidelines, please change the Findings heading to Results.

Response:
The heading has been changed

We hope that the manuscript now is acceptable for publication and we are looking forward to receiving your response.

Best regards,

Agneta Pettersson